Author's response to reviews

Title: 3,4-Dihydroxytoluene, a Metabolite of Rutin, Inhibits Inflammatory Responses in Lipopolysaccharide-Activated Macrophages by Reducing the Activation of NF-kappaB Signaling

Authors:

Yi-Lin Sophia Chen (a221865880@yahoo.com.tw)
Ya-Ping Chen (kii816@yahoo.com.tw)
Kuo-Feng Hua (kuofenghua@gmail.com)
Kang-Yi Su (suhy@ntu.edu.tw)

Version: 2 Date: 30 June 2013

Author's response to reviews: see over
Dear Editor of “BMC Complementary and Alternative Medicine BMC Complementary and Alternative Medicine”,

Attached please find the revised manuscript entitled “3,4-Dihydroxytoluene, a Metabolite of Rutin, Inhibits Inflammatory Responses in Lipopolysaccharide-Activated Macrophages by Reducing the Activation of NF-κB Signaling” which we would like to be considered for publication in “BMC Complementary and Alternative Medicine BMC Complementary and Alternative Medicine”. We revised the manuscript according to the editorial requirements and response to the comments point-by-point as below.

Requirement #1:
Please include Competing Interests, and Authors' Contributions sections in your manuscript.

Response: We included Competing Interests and Authors' Contributions sections in the revised manuscript.

Requirement #2: Abstract: please format your abstract according to the guidelines for authors.

Response: We changed the abstract into the correct format in the revised manuscript.

Thank you.

Sincerely,

Sophia

Associate Professor
Department of Biotechnology and Animal Science, National Ilan University, Taiwan